Author's response to reviews

Title: Neuropsychological Outcomes in Adults Commencing Highly Active Anti-Retroviral Treatment in South Africa: A Prospective Study

Authors:

John A Joska (John.Joska@uct.ac.za)
Jennifer Westgarth-Taylor (jenny@water-rhapsody.co.za)
Jackie Hoare (hoare.jax@googlemail.co.za)
Kevin G Thomas (Kevin.Thomas@uct.ac.za)
Robert Paul (paulro@umsl.edu)
Landon Myer (Landon.Myer@uct.ac.za)
Dan J Stein (Dan.Stein@uct.ac.za)

Version: 3 Date: 19 December 2011

Author's response to reviews: see over
To the editor: BMC Infectious Diseases

Title of paper: Neuropsychological Outcomes in Adults Commencing Highly Active Anti-Retroviral Treatment in South Africa: A Prospective Study

MS: 1358541805551110

Date of first submission: 13 May 2011
Date of re-submission: 27 October 2011
Date of second revision: 19 December 2011

Corresponding author: Dr John A. Joska

Address of corresponding author: Department of Psychiatry and Mental Health, University of Cape Town, Anzio Road Observatory, 7925. Cape Town, South Africa

Telephone: +27214042164
Fax: +27214488158
Email: John.Joska@uct.ac.za

We thank the reviewer for re-evaluating the above manuscript. The reviewer correctly noted that the sentence on the bottom of page 14 did not make grammatical sense. We have removed the word “were” from this sentence and made additional edits to this paragraph to ensure that the passage makes clear sense.

We trust that these revisions have improved the manuscript, and hope that you will consider publishing it.

We look forward to hearing from you.

Yours sincerely,

John A. Joska